. . Spring 2010 CSC/CPE 365: Database Systems Alexander Dekhtyar . . Lab 2: Database Creation

Due date: Thursday, April 15, midnight

Please note: While the lab deadline is Thursday midnight, we will start a new lab on Thursday, April 17.

Submission Instructions The Lab 2 handout outlines the exact specs for creating the archive lab2-<ilastname>.zip or lab2-<ilastname>.tar.gz that you will be submitting. It is important for you to follow those isntructions fully and use the presecribed directory structure and filenames.

Note: Your files should use no sql*plus spool commands. We will spool your output for you.

Once you created your submission archive, submit it using the following

handin command:

 handin dekhtyar-grader lab02-365 <file>

Good luck!